The Effect of Limited Search Ability on the Quality of Competitive Rent-Seeking Clubs⁺

By

Shmuel Nitzan¹ and **Kobi Kriesler²**

Abstract

A competitive rent-seeking club (CRSC) offers its members the chance of winning a prize (status, position, privilege) by being selected, typically, by a civil servant or a politician. The selector replaces in our setting the usual contest success function; instead of determining the winner on the basis of the club-members' efforts, he selects the winner on the basis of quality. This paper focuses on the effect of incomplete search of the selector on the efficiency of democratic self-governing and decentralized RSC's that control admittance to the club and its transparency, assuming that quality of their members is fixed. The incomplete search of the selector is assumed to take the simple form of fixed random sampling of the contestants - the members of the CRSC. Our results imply that, even when active rent-seeking expenditures are disregarded, the decisions of CRSC's regarding their composition and transparency tend to reduce quality and are therefore inefficient

Key words: competitive rent-seeking clubs, self governance, decentralized decisions, quality competition, selection by fixed random sampling, simple majority rule, endogenous membership determination, transparency, inefficiency, quality decay. **JEL Classification:** D70, D71, D72

[•]We are especially indebted to Arthur Fishman, Martin Kolmar, Nava Kahana and Klaus Schmidt for their insightful comments. The first author is grateful to the CES at the University of Munich, the Tinbergen Institute at the University of Amsterdam, the CODE at the Universitat Autonoma de Barcelona and the IAE in Barcelona for their hospitality which enabled the completion of this study.

¹ Department of Economics, Bar Ilan University, Ramat Gan, 52900 Israel, Corresponding author, Tel: +972 3 531 8345, Fax: +972 3 738 4034, e-mail: <u>nitzans@mail.biu.ac.il</u>

² Department of Economics, Bar Ilan University, Ramat Gan, 52900 Israel.

1. Introduction

Inefficiency in rent-seeking contests is usually due to the resources spent by the contestants in attempting to win a contested rent. In such contests, the rent seekers behave strategically and expend resources taking into account the contest success function; the function that determines their winning probability on the basis of their rent-seeking efforts, see for example, Baik (1999),(2008), Baik, Kim and Na (2001), Congleton, Hillman and Konrad (2008), Gradstein (1995), Hausken (2005), Konrad (2008), Nitzan (1994) and Nti (1997). The main objective of this study is to show that inefficiency in rent-seeking contests can also be due to quality competition under incomplete search of the agent who determines the winner of the contest - the selector, who is typically a bureaucrat or a politician. The incomplete search of the selector can be due to bounded rationality or to binding constraint on information gathering on the quality of the contestants. In our setting, the contestants are members of a competitive rent-seeking club (CRSC); they compete quality-wise on winning the prize, that is, on being chosen by the selector. Selection is an exclusive privilege of the club members. The quality of the club members is fixed, however, the extent of quality competition is determined by their decisions regarding the composition of the CRSC and regarding the transparency of their quality. The selector who tries to enhance his and the public well being has limited search ability or bounded rationality; he chooses the best (highest-quality) club member from the k members that he samples randomly. To clarify the issues arising in this setting, consider the following illustrative "drinking club" example.

Suppose that a group of n drinking buddies visit a bar. They like to drink beer and meet women. When an attractive woman meets the group, she has time to chat to only k out of the n men chosen at random and without replacement. Following this interaction, the woman will select the most attractive man from the k men she has spoken to. The men are exogenously ordered according to their quality (attractiveness), and they would like to maximize their probability of being selected by the women. A man succeeds in doing this if he has the opportunity to chat with the woman (which happens with probability k/n) and if there is no other man among the sampled group of size k who is more attractive. In this (politically incorrect) story, the n men are members of a competitive club and the woman is the selector. The story raises several questions: *First, is a man's probability of selection increased by the addition of an extra drinking man?*

The answer may be positive. For example, assume that k = n = 10. In this case the woman samples all the men and chooses the best one, who is ranked 1st. An unattractive entrant, when *k* remains fixed, reduces the selection probability of the 1st ranked man to (10/11) (there is a (1/11) probability of sampling 10 men that do not include the 1st-ranked man), increases the selection probability of the 2nd ranked man to (1/11) and leaves the selection probability of the remaining men unchanged at 0.

The entry of the (n+1)th agent has two effects on an incumbent: first, it reduces the probability of belonging to the sampled group from k/n to k/(n+1); second, it changes the incumbent's selection probability, provided that he is sampled. Entry of an inferior entrant increases this probability while entry of a superior entrant decreases it. Hence, the group may be divided into three subgroups: the most attractive men who are worse off by the increased competition, the most unattractive men who are indifferent (they face zero selection probability either way) and the remaining men who may be better or worse off due to appearance of the new competitor. The first result in section 2 provides necessary and sufficient conditions for an increase in an existing club member's selection probability due to increased competition (admission of a new club member).

Second, how many extra drinking men maximize an incumbent's probability of selection?

An incumbent's probability of selection may strictly increase due to the entry of a less attractive buddy. However, when too many new drinking buddies enter the bar, the incumbent's probability of selection decreases (when *n* goes to infinity the probability of selection for each man converges to zero). Hence, there must be a finite number of new and less attractive entrants that maximizes an incumbent's probability of selection. For example, it can be shown that if the sample size is 10, as in the former example, the best group size for the 1^{st} ranked man is 10 (no new club members), the best group size for the 2^{nd} ranked man is 20 (10 new low-quality members) and the best group size for the 3^{rd} ranked man is 30 (20 new low-quality entrants). The second result in section 2 establishes that the selection probability, as a function of the number of club members satisfies the single-peakedness property and , furthermore, for an incumbent ranked *r*, the optimal group size is *kr*, that is, his ranking multiplied by the sample size.

Third, assume that the drinking buddies decide who can join them, say, by some voting mechanism. How will this self-governing group evolve quality- wise and number-wise?

Suppose that the group, and in general, a democratic CRSC whose members expect a reward that hinges on their selection, applies a simple majority rule to make marginal entry decisions (about membership approval of single potential entrants) or to make a decision about the desirable size of the club (one time decision is made regarding the total number of entrants to the club). By the first result in section 3, when a democratic CRSC makes a separate decision regarding the admission of every potential entrant with a given quality, the outcome is either stagnation (no one is admitted to the club) or quality decay (only candidates whose quality is lower than the average quality of the club members are admitted). Such quality decay is inevitable when the group is big enough relative to the number of club members sampled by the selector. By the second result in section 3, when the democratic CRSC makes a nonmarginal decision regarding membership size, its quality usually deteriorates and is inversely related to the number of club members sampled by the selector.

Fourth, suppose that the number of drinking buddies is fixed, but they can affect the number k of sampled club members by revealing information on their quality. Will they disseminate such information?

Suppose that the amount of information about quality ranking of the club members positively affects k. The third result in section 3 clarifies that a democratic CRSC will not disseminate information on the quality of its members.

Fifth, can the women's situation be improved when admittance decisions or decisions on dissemination of quality information are made democratically by the men at the bar?

The answer is negative. The last result in section 3 establishes that the selector's situation cannot be improved if the decisions are made by a self-governing CRSC that applies a simple majority rule. In the illustrative drinking-men story, the drinking buddies will not reveal information on their quality and if a man is sufficiently attractive for the women (his quality is above the average quality of the man selected), he will not be allowed in to the bar.

Finally, assume that the drinking buddies determine the size and transparency of the club in a decentralized way; each man can admit as many new men as he *desires or disseminate any amount of information on the quality of the club members. What will be the size, transparency and quality of the group?*

By the two results in section 4, the equilibrium group size is increased and quality is reduced relative to the size and quality of a democratic self-governing CRSC and, under a fixed club size, the quality and transparency of a decentralized CRSC are increased.

Our basic observations are that certain members of a CRSC may, on one hand, welcome less able competitors and, on the other hand, object transparency regarding the quality of the club members. Consequently, when decisions on admitting new members and on transparency of the club members' quality are made democratically (applying a simple majority rule) or are decentralized, the decision of a CRSC regarding its membership and transparency tend to reduce its quality and, in turn, adversely affect welfare. As in the "drinking club" example described above, in our setting, the members of the CRSC have fixed quality, they wish to be selected and receive the reward of the quality contest and they control the composition of the CRSC and transparency of its members' quality. The selector who awards the "prize" has a special simple form of limited search ability or bounded rationality; he chooses the best (highest-quality) club member from the k members that he samples randomly. Section 2 presents the CRSC model and two preliminary results. In section 3 we explore the endogenous determination of membership size and of the dissemination of information on the quality of the club members in democratic self-governing CRSCs. Section 4 focuses on size and transparency determination in a decentralized CRSC. Section 5 contains three possible applications of our model. In the first examples the members of the CRSC are TV channels and doctors. In the third IO application, the "club" consists of brands produced by a multi-product firm and the selector is the consumer. The last section 6 contains a summary of our findings regarding the inefficiency of CRSC's and some concluding remarks. The proofs are relegated to an Appendix.

2. The setting and basic results

2(a). The competitive rent-seeking club (CRSC)

Consider an n-member CRSC, $n \ge 3$. The club members are exogenously strictly ranked according to their quality and they are assumed to know their ranking as well as the ranking of potential entrants to the club. The club is thus characterized by its

size *n* and by the strict ranking of its members. One of the club members is selected by an outside agent, henceforth, called the selector. Selection is an exclusive privilege of the club members. The club is competitive because its members compete (quality wise) on being selected. The club-members wish to be selected, in fact, to maximize their probability of selection, because selection results in some reward that is associated with the attainment of a certain position, status, job, prize or the selling of some good or (professional) service. The excludable commodity consumed by the club members is the chance of being selected and receiving the reward³. The consumed amount of this commodity may differ across club members because a club member's selection probability depends on his ranking, on *n* and on the behavior of the selector. Although quality of the club members is fixed, they can affect the extent of quality competition by their decision regarding the composition of the CRSC that determines their number, their ranking and the number of club members sampled by the selector.

The CRSC also offers its members the opportunity to consume at least one other excludable commodity, such as club reputation or, simply, income, which is assumed to be independent of the size of the club and the quality of its members. The existence of this additional commodity is needed in our setting to justify the incentive of prospective (existing) members with zero selection probability to join (stay in) the club. The independence of the benefit associated with the consumption of this additional club commodity enables us to disregard this extra benefit and focus just on a club member's selection probability while examining the effect of the size and quality of the club on the benefit of its members. Note that, even if the club does not offer the privilege of another excludable commodity, an agent with zero selection probability may have two other types of incentives to join or stay in the club. First, his selection probability may become positive due to the admission of lower quality candidates. Second, other club members may subsidize his admission to the club or his staying in it because his admission to (presence in) the club is advantageous to them.⁴

³On the notion of clubs, see Cornes and Sandler (1996, Chapter 3).

⁴ The admission of a low quality candidate to the club differently affects club members. Some benefit from his admission while the effect on other members may be harmful. In this case, some of the club members may have an incentive to subsidize such a new low quality candidate (like a pool insurance company that offers expensive insurance and is subsidized by the insurance group), while other members may have an incentive to lobby against the admission of such a member.

In our setting, the "reward" of a selected club member is assumed to be independent of the quality of the club. Consequently, the sole concern of a club member competing for selection is his selection probability. In some situations this assumption is not plausible. For example, a less attractive group of drinking buddies may eventually attract less attractive women to the bar. If the group quality and the agents' reward are positively correlated, an incumbent's tendency to prefer low-quality entrants may be offset by the reduced quality of the woman entering the (lowerquality) bar. Or consider the situation in a competitive academic club where selection of a member means that he receives a research grant. In this case, if faculty grants and quality are positively correlated, then current faculty members may prefer 'big shot' professors who may attract higher grants to the faculty, even if they reduce their selection probability.

2(b). The selector

In our stylized model, the selector is assumed to randomly sample *k* agents without replacement, $1 \le k \le n$, and choose the highest-ranked agent from this sample⁵. Hence, the environment of the competitive club and the selector is characterized by the club size *n*, the strict ranking of the club members and the selector's fixed sample size *k*, which can be interpreted as effective search constraint or as the extent of bounded rationality of the selector. The outcome of the competition among the club members on the expected reward depends therefore on the quality of the members which determines their ranking, on *n* and on *k*. In fact, in our setting the relevant contest success function relates the winning probability of a club member to his and the other members' quality. This function depends therefore on his ranking *r*, on *n* and on *k*. The explicit form of the function is given by equation (1) in the Appendix. Specifically, the probability that a selector chooses club member with ranking *r*, when *k* members are sampled, is:

(1)
$$\frac{(n-r)!(n-k)!k}{(n-r-k+1)!n!}$$

⁵ On models of bounded rationality and, in particular, on random choice functions, see Rubinstein (1998) and Rubinstein and Salant (2006).

Recall that in our setting, even-though the quality of the club members is assumed to be exogenously given⁶, they can still affect competition on the reward by admitting new members to the club or by reducing the size of the club and by affecting the selector's fixed sample size k.

Note that in our simple model the selected subset of club members is not chosen on the grounds of quality signals collected by the selector or offered by the club members. The k-member subset is chosen randomly and k is assumed to be fixed and independent of the size or the quality of the club members. Although, for the sake of simplicity, k is considered as a primitive, it may be the outcome of a maximization process. The selector could care about the relative and not the absolute expected quality of the chosen alternative. For example, if the selector wants to choose an agent that is ranked, on average, in the top 50%, she has to sample only one agent regardless of the sample size. If the selector wants to choose an agent that is ranked, on average, in the top 66%, she has to sample two agents regardless of the sample size. It is easy to verify⁷ that if the selector wishes to choose an agent who is ranked, on average, in the top α %, by sampling a minimal number of agents, she has to sample $\alpha/(1-\alpha)$ agents, regardless of the sample size. The simple assumptions on the specific procedure adopted by the selector enable the derivation of our two basic preliminary results.

2(c). The effect of marginal admittance on the probability of selection

Suppose that the *n* incumbents, the existing club members, are augmented by a single entrant to the club. Note that in this simple setting, increased competition, that is, admission of a new member to the club, may strictly increase the selection probability of some incumbents. For example, suppose that k = n. In this case the selector samples all the club members and chooses the best member, who is ranked 1st. A new inferior member, when *k* remains fixed, reduces the selection probability of the 1st ranked member to (n/(n+1)) (there is a (1/(n+1)) probability of sampling *n* agents that

⁶ In our setting quality externalities are disregarded. Therefore, the quality of existing members is not affected by the admittance of new members.

⁷ To formally prove this assertion, note that the expected ranking of the agent the selector chooses is equal to ((n+1)/(k+1)). The expected relative ranking is therefore equal to ((n+1)/(k+1)n). When the selector insists on a relative ranking that does not exceed α , the minimal *k* that secures this α is equal to $k^*=(\alpha/(1-\alpha))+(1/n(1-\alpha))$. And since $(1/n(1-\alpha)) < 1$, if the selector wants to secure α and do it by sampling the smallest number of agents, then indeed his optimal sample size is robust, that is, k^* is independent of *n*.

do not include the 1st-ranked agent), increases the selection probability of the 2nd ranked agent to (1/(n+1)) and leaves the selection probability of the remaining agents unchanged at 0.

More formally, let r^* be the ranking from the top of the new entrant, $r^* \in \{1, 2, ..., n+1\}$. Note that the admittance of this agent positions him in the r^* place while decreasing (adversely affecting) the ranking of all the agents of lower quality by one position. The following proposition provides the necessary and sufficient conditions for an increase in an existing agent's selection probability due to increased competition (admission of a new member to the club).

Proposition 1: Let *r* be the ranking of member *x*. A new entrant, ranked r^* , increases the selection probability of member *x* iff:

$$\min[(n-k+3), r^*] > r > \frac{n+1}{k}.$$

Proof: See Appendix.

A club member's probability of selection is equal to the probability that this member is sampled, multiplied by his conditional probability of selection (the probability that the member is chosen if he is sampled). A new entrant decreases the probability that the club member will be sampled. The conditional probability of selection depends on the quality of the new entrant: the admittance of an inferior agent increases this probability while the admittance of a superior agent decreases it. Hence, a new entrant can only increase the probability of members who are superior to the new member. By Proposition 1, a new inferior entrant increases the selection probability of other club members who are not ranked very high (they do not belong to the top (n+1)/k)quality members) or very low (such that their selection probability is 0 before and after admittance). The following graph illustrates Proposition 1 (for the case where r*<(n-k+3)):



Note that in the special case where the quality of the prospective entrant is inferior to the quality of all the existing club members, that is, $r^* = n+1$, by Proposition 1, we obtain:

Corollary 1: Let r be the ranking of club member x. A new inferior entrant increases the selection probability of member x iff:

$$(n-k+3) > r > \frac{n+1}{k}.$$

The following graph illustrates Corollary 1:



The former example, where k = n, is of course consistent with Corollary 1, according to which, in such a case only the selection probability of the second-best member, the individual ranked in the second position (r = 2), increases, because 3 > r > (1+(1/n)).

To sum up, by Proposition 1 (and its corollary that focuses on the special case where $r^* = n+1$), increased competition adversely affects club members who are inferior to the new entrant whose ranking is lower than r^* , the ranking of the new entrant, as well as some high-quality members whose ranking is higher than r^* , and positively affects the remaining club members (if there are any). This latter effect is essentially due to *k* being smaller than (n+1).

2(d). Club members' optimal membership size

An incumbent's probability of selection may strictly increase due to the admission of a low-quality member to the club. The admission of too many low-quality members to the club decreases an incumbent's probability of selection (when n goes to infinity the probability of selection of each agent converges to zero). Hence, there must be a finite number of new and less attractive entrants that maximizes an incumbent's probability of selection. The following result specifies the most preferred number of club

members from the viewpoint of a member ranked r, given that the selector samples k agents.

Proposition 2: Suppose that *k* club members are sampled. The maximal selection probability of a member ranked *r* is obtained when the number of club members is equal to $n_r = kr$.

Proof: See Appendix.

For example, assume that k = 5. Figure I illustrates the selection probability (in %) of a member ranked 3^{rd} , r = 3, as a function of the number of club members.



Figure I

Note that when 5 members are sampled and there are 5 or 6 members in the club, the 3^{rd} ranked member cannot be chosen (the 3^{rd} ranked member cannot be the best member in the sample). The maximal selection probability of a member ranked 3^{rd} is 16.5% and it is achieved when the number of club members is equal to 15 (by Proposition 3, $n_r = kr = 3*5$).

3. Democratic self - governing CRSC's

3(a). Membership determination ⁸

⁸ In the standard rent-seeking literature, the issue of admission of contestants is also examined, albeit usually from the point of view of the contest designer and not that of the group of contestants. A useful survey of this literature can be found in Konrad (2008, section 3.3).

In a self-governing CRSC, the current members are free to determine, one way or another, whether or not to admit a potential candidate to the club. Suppose that approval of admittance of a new agent is determined democratically by applying a simple majority rule; that is, the decision to approve admission of a new member to the club is made only if it gets the support (vote) of a simple majority. Recall that, by assumption, the incumbents know their ranking as well as the ranking of the potential entrant. An existing member votes in favor of admission of a new member provided that such admission strictly increases his selection probability. The new member is admitted to the club if at least half of the existing club members approve its admission.

A CRSC is called *stable* if the existing members prevent any admittance, regardless of the prospective entrant's quality. A *decaying CRSC* is an unstable club that is vulnerable to decay; that is, it is vulnerable to an endogenous decline in the expected quality of its members. In other words, the existing members in a decaying club approve only prospective entrants that reduce the average quality of the club members. The following proposition describes the outcome of marginal admittance decisions in a self-governing CRSC that applies a simple majority rule and provides a sufficient condition for decay.

Proposition 3: A self-governing CRSC that applies a simple majority rule to admit members is either stable or decaying. Assuming that the selector samples $k \ge 3$ agents, a sufficient condition for decay is that $k \le 0.5n$. **Proof**: See Appendix.

That is, self-governance based on the use of a simple majority rule results either in stability (stagnation) or in quality decay. By Proposition 1, a potential entrant of above-average quality is rejected by all club members of less than average quality and by some top-quality members; hence a majority of the members never approves the admission of such a candidate. Only potential members of less than average quality can be approved and this happens when the sufficient condition is satisfied. Notice that if *k* is sufficiently high relative to *n*, then the admission of a prospective entrant, regardless of quality, is never approved and the club is stable. For example, suppose that k = n. By Proposition 1, the majority of the club members, that is, the 1st ranked

member as well as the n-2 = k-2 low-quality members reject the entrant, and only the 2nd ranked member approves it. If k is sufficiently small relative to n, then the admission of an inferior prospective entrant is approved. For example, suppose that k = 3 and n = 100. By Proposition 1, the 33 top-quality club members as well as the 100th ranked member reject the entrant, while the majority of the club members (66 of them) approve it. By Proposition 3, given n, the threshold sampling size is k = 0.5n. That is, if half or fewer of the existing members are sampled⁹, then the club decays. Put differently, given k, if the existing number of club members exceeds a certain threshold, $n \ge 2k$, then self-governance based on the simple majority rule necessarily results in quality decay. Suppose, for example, that k = 3 and that a new inferior agent proposes his candidacy to the club. Table I specifies the number of club members that vote in favor of his admission. In this case, if the number of the existing members is equal to or exceeds 6, then admission of the inferior agent is approved and the club decays. When the number of the existing club members is smaller than 6, the club is stable or decays.

⁹ This observation is not valid in the extreme cases where k = 1 or 2 and *n* is odd. By proposition 1, in the former case all the existing members reject entry. In the latter case, a majority of the members reject entry.

Size of	<i>n</i> +1	Members who vote against admission of			Members	who	Admittan
the club	k	the new candidate			vote in favor of		ce
		Top-quality		Low-	the	new	approved
				quality	candidate		
		Decreased	Unchanged	Unchanged	Number ¹⁰	%	
		selection	selection	selection			
		probability	probability	probability			
		if	if				
		$r < \frac{n+1}{2}$	$r = \frac{n+1}{2}$				
		k k	, k				
3	1.3	1		1	1	33%	No
4	1.7	1		1	2	50%	Yes
5	2	1	1	1	2	40%	No
6	2.3	2		1	3	50%	Yes
7	2.7	2		1	4	57%	Yes
8	3	2	1	1	4	50%	Yes
9	3.3	3		1	5	56%	Yes
10	3.7	3		1	6	60%	Yes
11	4	3	1	1	6	55%	Yes
12	4.3	4		1	7	58%	Yes
13	4.7	4		1	8	62%	Yes
14	5	4	1	1	8	57%	Yes
15	5.3	5		1	9	60%	Yes

Table I

Note that the basic model can easily be extended to a multi-period model. Suppose that there are *j* periods such that in each period t_i , $i \in \{1, 2, ..., j\}$, there is a prospective entrant of ranking r_i . This ranking is known to the other club members. Furthermore, suppose that all the club members have the same discount factor δ , $0 \le \delta \le 1$, and that each member maximizes his discounted selection probability. For a small enough δ

¹⁰ Note that the number of members who vote in favor of the new entrant (column 6), is derived by subtracting the number of members who vote against the new entrant (columns 3-5) from the total number of club members (column 1).

(such as in the myopic case where $\delta = 0$), the results for each period are identical to those obtained in the one-period model. In such a case, the self-governing club would demonstrate either stability or continuous growth in membership, accompanied by an endogenous decline in the club's expected quality.

Suppose now that admittance decisions are not made on the margin; that is, the issue is no longer whether to admit or reject a potential entrant to the CRSC, but to determine the number of club members. The equilibrium club membership naturally hinges on the preferences of the existing members regarding the size of the club. Under the commonly used simple majority rule, the chosen club membership size is the Condorcet winner; namely, the club-size proposal that defeats by a simple majority any other proposal. If such a winner does not exist, then the status quo n is the chosen club size. To apply Proposition 2 in determining the preferred club size for every voter, we preserve the ranking r of every club member by assuming that potential entrants are inferior low-quality agents (the ranking of a potential entrant r^* is lower than n). By Proposition 2, we obtain the following result:

Proposition 4: Suppose that the number of existing club members *n* is odd. Then the equilibrium size of a self-governing CRSC that applies a simple majority rule is k(n+1)/2. That is, when $k \ge 3$ the *n* existing club members approve the admission of $\{k(n+1)/2 - n\}$ new inferior low-quality members.

Proof: See Appendix

Thus, in a democratic self-governing CRSC that applies a simple majority rule, the most preferred club size, ((k(n+1))/2), of the median-quality member, the member ranked (n+1)/2)), is the equilibrium club membership (the Condorcet winning alternative), see Gans and Smart (1996). This means that the equilibrium club membership increases with the existing fixed k. In the extreme case of limited search ability or of the selector's bounded rationality, k=1, the median-quality club member actually prefers to reduce the club size to (n+1)/2, dismissing all members whose quality is lower than his quality. If such dismissal is feasible, the median voter becomes the worst-quality club member and, in turn, the quality of the club is increased. In the extreme case of unconstrained search ability or r unconstrained rationality, k=n, the equilibrium size is (n(n+1))/2 and the resulting deterioration in the club quality is maximal. Notice that club quality is inversely related to the fixed

sample size k. In the previous case of marginal admission decisions, unbounded search ability of the consumer results in preservation of the quality of the club. In contrast, in the current case of a non-marginal decision on the size of the club, the initial unbounded search of the selector results in bounded search because the size of the club changes from n to n(n+1)/2 and this maximal club size results in the largest reduction in quality.

Finally note that the self-governing CRSC may apply alternative democratic rules to determine its size, marginally or non-marginally. One extreme such rule is the hierarchy rule (see Sah & Stiglitz, 1986). Under hierarchy, a potential entrant is approved if all the existing club members vote for his admission. Otherwise he is rejected. Clearly, by Proposition 1, under hierarchy, no candidate is admitted to the club and therefore the existing quality is maintained. In a non-dichotomous setting, under hierarchy, a particular club size is the choice of the self-governing club, if it is supported by all the existing club members who take part in the decision. If such unanimous support cannot be secured, then the status quo n is the chosen club size. Propositions 1 and 2 imply that under hierarchy, the self-governing CRSC chooses to prohibit entry. That is, again, under hierarchy n is the chosen size of the club because the highest-quality member who objects entry to the club is decisive. Hence, hierarchy is superior to the simple majority rule in terms of its effect on club quality.

3(b). Information disclosure affecting the number of sampled club members

Suppose that the competitive CRSC has a data base that can be useful in revealing the relative quality of its members (C.V's of club members, information on their past performance, etc.). The control of such information is typically carried out by decisions on the nature of the data base of the club and the accessibility of that data base. Assume that dissemination of information on the quality of the club members positively affects the actual fixed sample size k, that is, the more information the competitive club reveals, the easier it becomes to sample more members and, consequently, the selector increases k^{11} . In the democratic self-governing CRSC, the decision on how much information to disseminate is also reached by applying the simple majority rule. In such a case the equilibrium decision of the club is to avoid any dissemination of information on the quality of its members. That is,

¹¹In such a case, the "contest success function" in our model is affected by the decision of the CRSC on information disclosure.

Proposition 5: Suppose that the dissemination of information on the quality of the club members positively affects the selector's fixed sample size k. Then a self-governing CRSC that applies a simple majority rule will not disseminate information on its quality.

Proof: See Appendix.

When k = 1, each member has a selection probability of 1/n. When k = 2, the selection probability of the median member is 1/n, whereas the selection probability of higher (lower) quality members is higher (lower) than 1/n. When $k \ge 3$, the selection probability of a majority of the members is less than 1/n, hence the majority of the club members strictly prefer a small sample size k (k = 1 or k = 2) and given that dissemination of information on the quality of the club members positively affects the actual fixed sample size k, the equilibrium decision of the CRSC is to avoid any dissemination of information on the quality of its members.

3(c). The selector's welfare

The selector chooses the best sampled club member. In our probabilistic setting, the selector's preferences are represented by the expected quality of the selected club member. Note that this is an ordinal measure because, given the quality of the existing club members, we assume that a member's quality is represented by his ranking. In particular, the ranking of a potential entrant is represented by his ranking relative to the ranking of the existing club members. The use of such an ordinal measure implies that the selector is only concerned about how well he can do, given the existing quality of the club.¹² As we have seen, the democratic decisions of a competitive club about admittance of new members adversely affect its quality.¹³ Our last result establishes that the expected welfare of the selector cannot be improved when the self-governing club resorts to a simple majority rule.

¹² The application of the measure is therefore not plausible for inter-club welfare comparisons of the selector because, obviously, selection of the best-quality member in one club, is not necessarily better than the selection of a medium-quality member of another different club.

¹³ This conclusion is valid for any voting rule like hierarchy that is positively responsive to the club members' votes.

Proposition 6: Suppose that a competitive self-governing CRSC applies a simple majority rule when making marginal or non-marginal admittance decisions or when making a decision on disclosure of information about the quality of its members. Then the CRSC never makes a decision that is advantageous to the selector. **Proof:** See Appendix.

Basically, if a new potential entrant has a sufficiently high quality (higher than (n+1)/(k+1)), so that the selector benefits from his entry, all the members of higher quality vote against his admission (they belong to the (n+1)/k top-quality members of the club) and so do the members of lower quality. Since all the existing members reject such a potential entrant, no democratic voting rule applied by the self-governing club can benefit the selector.

Disclosure of information on the quality of the club members may be advantageous to the selector. However, the democratic club prefers not to disseminate such information. Proposition 6 thus implies that the selector can be "saved" only by external regulation of club membership and of transparency of club-members' quality.

4. Decentralized CRSC's

In a decentralized CRSC, every member can approve a candidate to the club. In other words, a candidate ranked r^* is admitted to the group iff there exists r, such that $\min[(n-k+3), r^*] > r > \frac{n+1}{k}$. In such a club, admittance of a high-quality candidate (possibly, with above-average quality) with ranking $r^* > [(n+1)/k]+1$, is therefore approved. Although a candidate with above-average quality can be admitted to the club, in general, the effect of marginal entry on quality is ambiguous. Sufficient conditions for quality decay of a decentralized CRSC can, however, be provided. For example, it can be verified that if the ranking r^* of a prospective entrant is uniformly distributed over the possible rankings 1 to n, then the quality of an approved entrant is lower than the average quality, i.e., in such a case the decentralized CRSC decays. Notice that, by Proposition 3, in the dichotomous setting where admittance of a candidate is approved or rejected, the application of the simple majority rule, that can never improve quality, is more harmful to the club's quality.

Consider now the non-dichotomous setting, assuming that the club makes decisions in a decentralized way. In this case, every member i, i = 1,...,n, makes a

proposal e_i regarding the number of candidates that should be admitted to the club, and every proposal is approved, which means that the decentralized CRSC approves admittance of $e=e_1+e_2+\ldots+e_n$ new members. The chosen club size is therefore (n+e). Note that we now allow strategic decisions by the existing club members. That is, a club member is not assumed to behave sincerely (make a sincere proposal). To apply Proposition 2 in determining the preferred club size for every member, we preserve the ranking r of every club member by assuming that potential entrants are inferior low-quality agents (the ranking of a potential entrant r^* is lower than n). By Proposition 2, we obtain the last two results:

Proposition 7: In a decentralized CRSC, the (Nash) equilibrium size of the club is nk. That is; the existing n members approve the admission of (k-1)n inferior low-quality candidates

Proof: See Appendix.

In a decentralized CRSC, when the size of the club is strategically determined by the existing members analogously to the voluntary provision of a public good, the lowestquality member is actually decisive; the equilibrium number of club members maximizes his selection probability $(n_n k = nk)$ is the most preferred club size of the lowest-quality member, the agent ranked n).¹⁴ Note that, in contrast to the superiority of a decentralized club to the democratic self-governing club that applies a simple majority rule in the context of dichotomous (marginal) admittance decisions, in the current setting of non-marginal club size determination, the democratic club is preferable to the decentralized club. The reason for this reversal in the effect on quality under the two types of clubs is the following: Under marginal admittance decisive club than in the decentralized club. In the self-governing CRSC, the ranking of the decisive club member is d = [(n+1)/k] + [(n+1)/2],¹⁵ whereas in a decentralized CRSC his ranking is [(n+1)/k] + 1. Consequently, admittance of a high-quality candidate (possibly with

¹⁴ In fact, it can be shown that the same result is valid, even if the quality of potential entrants is not restricted to being inferior.

¹⁵ By Proposition 1, the member ranked (n+1)/k and the higher-quality members reject the potential new entrant. To be approved by a simple majority, the potential entrant must secure the support of at least (n+1)/2 of the highest quality members amongst the remaining members. This means that the ranking of the decisive club member; that is, the highest quality member that secures a simple majority, is equal to [(n+1)/k]+[(n+1)/2].

above-average quality) of a ranking r^* , $d > r^* > [(n+1)/k]+1$, is prevented under the simple majority rule, but approved under decentralized decisions. Under non-marginal admittance decisions, the decisive club member is of higher quality under the simple majority rule than under decentralized decisions. In the former case, the ranking of the decisive club member is (n+1)/2, the decisive member is the median voter, whereas under the latter case, the ranking of the decisive member is *n*, the decisive member is the lowest-quality member. Consequently, more inferior low-quality candidates enter the club under decentralized decision making than under the simple majority rule, $\{kn - n\} > \{k(n+1)/2 - n\}$.

The last result implied by Proposition 2 relates to quality transparency in a decentralized CRSC.

Proposition 8: Suppose that the members of a fixed-size CRSC have equal ability to disseminate information on the quality of the club members and that this information positively affects the selector's fixed sample size k. Then in a decentralized CRSC, the (Nash) equilibrium of disseminated information on the club's members' quality is maximal.

Proof: See Appendix.

In a fixed-size decentralized CRSC whose members can disseminate information on the club members' quality, the highest-quality member, the member ranked 1, is actually decisive. This club member is interested in maximizing the selector's sample size k. In fact, his maximal selection probability of 1 is obtained when k=n. His maximal feasible selection probability is obtained when he disseminates as much information as he can on the quality of the club members. By Proposition 8, when the size of the club is fixed, the decentralized club is more transparent than the democratic self-governing club and, in turn, its quality is higher.

5. Applications

5(a). The quality of TV channels: The justification of external regulation Suppose that the competitive club consists of TV channels that seek selection by potential watchers to improve their rating and that consumers of TV make their best channel choice after randomly sampling k out of the existing n channels.¹⁶ To fit our CRSC setting, suppose that the government awards a certain privilege to the most popular channel. In this case our results imply that the expected quality of TV services can only deteriorate in a decentralized setting, where the existing club members, the operating channels, can establish new channels, that is, co-opt new club members while having in mind just their probability of winning the contest reward. The same conclusion is valid in a democratic setting where the operating channels are viewed as a self governing club. This assumption is realistic in a democratic environment where the decisions of the regulator of the TV industry directly reflect the interests of the existing channels and his decisions are made by applying a rule which is equivalent to the simple majority rule. In this case we also expect quality decay of the TV services. Our model can therefore provide a simple straightforward rationalization for quality regulation of the TV channels club.

5(b). Disclosure of information on medical expertise

Information on medical expertise in different fields is almost never disseminated by the relevant professional organizations. In the United States, for instance, even the questionable individual ranking of heart surgeons according to success in bypass coronary has only appeared since 1991 in New York State and the only other state that reveals the ranking is Pennsylvania, Dyer (2003). This observation is rationalized by Proposition 5, assuming that medical professional organizations are democratic self-governing competitive clubs whose members wish to be approved or selected for offering their services to government agencies or medical insurance companies by government officials who randomly sample a fixed number of the club members. The doctors in this example have in mind the probability of winning the reward associated with selection to offer their professional services. The selectors in this context, the bureaucrats, are typically very much concerned about the quality of the chosen club member - the medical expert. Even when the bureaucrat's decision is affected by price, information on quality is crucial. When price is not an issue, for example when all of the payment for the medical service is borne by the medical insurance company, the quality of the expertise is the sole and main issue. The bureaucrats face a most difficult selection decision and the lack of transparency regarding the quality of the

¹⁶ On the irrationality of consumers of TV channels, see Benesch, Frey and Stutzer (2006).

available medical experts is certainly not to their advantage (Proposition 6). The government can play a useful role in correcting the failure of the democratic competitive club of medical professionals. The objective difficulties of ranking medical experts, together with the fact that the politically powerful medical profession has no interest in disclosing this information, may explain the common disregard of the government in making available transparent medical expertise, especially when the medical services are offered directly to the public. Interestingly, in such cases, alternative non-profit organizations do help the patients. For example, in Israel, two ultra orthodox charity organizations disclose information on medical expertise. These organizations build and maintain a medical data base and grant accessibility to this data base to anybody free of charge. The outstanding popularity and efficiency of these organizations attests to their ability to identify and cope with the lack of quality transparency in the medical profession and its detrimental welfare implications.¹⁷

5©.. A new justification of branding

The last possible application that we propose is less directly related to our CRSC setting, but rather to the justification of branding, an issue that has been of much concern in the IO literature. However, this application makes a direct use of the results derived in our study. Suppose that a firm can produce different brands and, in particular, low-quality brands of its main product. In such a case the variety of brands can be conceived as a competitive club – the club of brands the firms wish to sell to consumers who make their best-brand purchase decision after randomly sampling *k* of the *n* existing brands. Notice that since the brands are ranked only according to quality, their price is disregarded, as in fixed price stores, or the brands quality takes into account all aspects, including their different prices. In this case, the sale probability of a leading brand may increase when new low-quality brands (or new firms that produce them) are introduced. This implies that a firm may have an incentive to introduce such low-quality brands, even if such brands have very small or zero sale probability. When the firm can produce low-quality brands at zero cost,¹⁸ by

¹⁷ The contribution of one of these organizations to the welfare of the Israeli society was recognized in 1997 when Rabbi Avraham Elimelech Firer, the chairman and founder in 1979 of "Ezra Le Marpeh" Association has received the Israel Award, the most prestigious civilian award granted to persons who contribute to social life in Israel.

¹⁸ This is a plausible assumption when products are "virtual", like some unattractive products in brochures that are never meant to be sold, or when the products are versions of an existing product, possibly a damaged product, that just carry new labeling.

Proposition 2, the optimal number of such brands; that is, the number that maximizes the sale probability of the leading brand, is equal to (*kr-n*). For example, if, there are already 13 brands in the market, n = 13, and, k = 5, a firm ranked 3rd, introduces two low-quality brands in order to maximize the sale probability of its leading brand. By Proposition 7, if every firm can produce any number of low-quality brands at zero cost, then under decentralized branding decisions, the (Nash) equilibrium industry size is *nk*. That is; the lowest-quality firm introduces (*k*-1)*n* new low-quality brands. By Proposition 4, when the number of existing firms *n* is odd and every firm can produce any number of low-quality brands at zero cost, the equilibrium industry size in a self-regulating industry that applies a simple majority rule is k(n+1)/2. That is, the *n* existing firms introduce {n(k-2)+k/2} new low-quality brands.

The sale of a new product by a multi-product firm can be warranted because it may attract new consumers to the market, it may cause consumers who purchase products from competing firms to switch to the new product or it may "cannibalize" the firm's own line of products. According to the existing literature, the firm's incentives for introducing a new product are basically based on the existence of a tradeoff between the extra profit from the new product and the adverse effect of its introduction on the firm's profits from the existing line of products. In particular, new and inferior products (like the IBM Laser Printer that prints 5 pages per minute, that was created by adding a speed-limiting chip to IBM's Laser Printer of 10 pages p.m., Deneckere & Mcafee, (1996)), may be introduced for various reasons intended to exploit the heterogeneity in the consumers' preferences or income, to deter prospective entrants or to "fight" actual new entrants. Unlike any of the existing models, by Propositions 2 and 3, the sale probability of the leading brand in our setting may increase when new low-quality brands (or new firms that produce them) are introduced. This implies that, due to k being relatively small (the sufficiently bounded rationality of consumers) or n being relatively large (the fact that the existing size of the industry is sufficiently large relative to the number of firms sampled by the consumer), and not due to the heterogeneity of the consumers' income or preferences, a firm may have an incentive to introduce such low-quality brands, even if such brands have very small or zero sale probability. When there are no potential entrants and the size of the industry *n* cannot be changed by establishing new firms or by producing new brands, the existing firms may affect the sample size k, for example, by controlling consumers' information on the firms (allowing advertising, requiring

that the firms' qualifications are known, etc.). In such a case, by Proposition 8, decentralized decisions by the firms will result in maximal disclosure of information regarding the quality ranking of the firms. In contrast, by Proposition 5, a collective democratic decision by the firms will result in no disclosure of information regarding the quality ranking of the firms.

6. Summary and concluding remarks

In this study we have focused on CRSC's whose members are of heterogeneous fixed quality and on a selector (bureaucrat or politician), who does not know the members' quality, randomly sampling a fixed number of club members and then choosing the best-sampled member. Selection of a member means that he receives the reward associated with a certain position, status, job or prize. Under the above assumptions, an existing club member may become better off due to enhanced competition; that is, the admission of a new member to the club. More specifically, by Proposition 1, the newly admitted member may harm the top-quality members, harm or not change the situation of the low-quality members and benefit the medium-quality club members. We then established in Proposition 2 that, given the fixed sample size k and the ranking of a club member r, that member maximizes his selection probability when $n_r = k*r$; that is, when the number of club members is equal to the fixed sample size multiplied by his ranking.

The analysis then focused on the implications of the model in the context of a self-governing CRSC; a club in which the existing members control entry, marginally or non-marginally, and in turn, the quality of the CRSC or control disclosure of information on the quality of the club member. By Proposition 3, under marginal admittance decisions, if the self-governing CRSC resorts to simple majority rule, then the outcome is either stability or continued growth in membership, which is accompanied by a decline in the quality of the club. A sufficient condition for such quality decay is that $k \le 0.5n$; that is, the existing club size is sufficiently large relative to the number of club members sampled by the selector.¹⁹

We derived the non-marginal equilibrium club size when the democratic selfgoverning club resorts to simple majority rule. We have shown in Proposition 4 that under a simple majority rule, when $k \ge 3$, the club decides to increase its size. Under

¹⁹ We showed that stability of a self-governing club is also secured under hierarchy, but not necessarily in a decentralized club.

the equilibrium club membership, the average quality is reduced. The reduction in quality is maximal when initially k=n. The reason is that in such a case of initial unbounded search, the club admits the largest number of new low-quality members. When the size of the club is fixed, it may affect the fixed number of club members sampled by the selector by disclosing information on the quality of the members. By Proposition 5, in such a case the democratic CRSC refrains from revealing such information.

We then focused on the welfare of the selector that in our case is equivalent to that of society. In our setting, the selector's preferences are represented by the expected quality of the chosen agent. By Proposition 6, if a new potential entrant has a sufficiently high quality that is beneficial to the selector, then he will not be admitted to the club. Furthermore, any information regarding the quality of the club members that can increase the expected quality of the selected club member will not be released. That is, the democratic CRSC has an incentive to prevent entry and quality transparency that are welfare enhancing.

Our model predicts some basic undesirable features of self-governing CRSC's. The focus is not on the standard aggregate rent-seeking efforts of the CRSC or on the efforts of its members to compete directly, by affecting their quality or by sending signals (to the selector) on their existing quality. Rather, given the limited search ability of the selector or his bounded rationality, we examine the effect of the decisions of the CRSC regarding its composition and regarding its transparency, respectively, on the competitive environment and on the assumed contest success function. The combination of selfish rent seekers who are interested in being chosen by the selector (civil servants, politicians), fixed random sampling of these rent seekers by the selector, and the fact that the self-governing CRSC resorts to simple majority rule, can only result in unchanged or deteriorating quality of the club and of the expected quality of the selected club member.²⁰ This is the inefficiency obtained in our quality contest which is due to the effective club members' incentives to welcome less able competitors and to object transparency regarding quality. The effectiveness of the incentives is rendered possible due the reliance of the self-

²⁰ Note that even in the standard rent-seeking literature, the source of inefficiency can be due to the reduced probability that the most efficient contestant is chosen rather than to increased (wasteful) rent-seeking expenditures. This was pointed out, for example, by Ellingsen (1991) and Fabella (1995) and, more recently, by Amegashie (2000).

governing CRSC on democratic decision making, viz., the application of the simple majority rule.

When the size and transparency of the club are determined in a decentralized way, the most active members play a dominant role. Specifically, the least qualified and the most qualified members play a decisive role in determining, respectively, the size and the transparency of the club. By Proposition 7, the size of the decentralized CRSC is larger than the size of the democratic self-governing CRSC and its quality is lower. The superior performance of the democratic club is due to the decisiveness of the less active median-quality member, relative to the more active harmful role played by the lowest-quality member in a decentralized CRSC. By Proposition 8, when the size of the club is fixed, the decentralized CRSC is more transparent than the self-governing CRSC and its quality is higher. The advantageous functioning of the decentralized CRSC is due to the decisiveness of the more active highest-quality club member, who disseminates as much information as he can, relative to the less active detrimental role played by the decisive median-quality member in a democratic self-governing club.

From the welfare point of view, the outcome of a self governing CRSC is inefficient because it results in more inferior members and less information on the quality of the club members (actually no information). Transforming such a club into a decentralized CRSC changes the outcome; the number of inferior members is increased, however, more information is disclosed. This means that the efficiency effect of such an institutional transformation is ambiguous; the outcome does not necessarily become more efficient.

Our analysis of CRSC's has focused on democratic self-governing clubs and decentralized clubs that have two main simple features. First, the benefit of the club members from being chosen by the selector and from consuming other excludable club goods are independent of the size and the quality of the club. Second, the club faces a particular form of limited search ability or bounded rationality of the selector who samples a fixed number of club members before making his choice. Alternative assumptions regarding the benefit of the club members and the behavior of the selector can be made. For example, the benefit from selection can be positively related to the quality of the club. The benefit from consuming other excludable club goods, such as reputation, can be inversely related to the size of the club and positively related to its size. The sample size of the selector need not be fixed; it can

be inversely related to the quality of the club. And the selector need not sample club members randomly. Examining the implications of these alternative, perhaps more realistic and less stylized, assumptions on the size, transparency and quality of CRS's is certainly worth pursuing. However, this task is beyond the scope of the current paper and is left for future research.

The reward of the selected club member can be associated with the selling of some good or (professional) service that he and the other members attempt to sell to the government or to the public. In such a case, our findings can shed some light on the issue of whether industry regulation is warranted. In particular, they can provide rationalization for the regulation of industries that satisfy the basic features of our CRSC, as illustrated in the example of the TV channels. We can also apply our model to shed light on the observed lack of transparency in certain professions, such as the medical profession example we have briefly discussed and on the reason that charity organizations have to disclose information, e.g., on medical expertise. In the latter case, the reason seems to be at least partly a response of the charity organizations to their realization that the democratic competitive club of medical professionals and the government have no interest in disclosing this information. Finally, we have presented a possible application of our model to branding, an issue that is of much concern in the IO literature. In this application we have conceived the variety of brands produced by a firm as a decentralized or a self-governing competitive "club of products or brands". In this case, our model can provide interesting rationalization of versioning, and the endogenous determination of club membership could relate to the choice of the extent of versioning or branding by a multi-product firm.

Appendix

Proposition 1: Let *r* be the ranking of member *x*. A new entrant, ranked r^* , increases the selection probability of member *x* iff:

$$\min[(n-k+3), r^*] > r > \frac{n+1}{k}.$$

Proof:

The probability that a selector chooses club member *x*, when *k* members are sampled, is:

(1)
$$\frac{(n-r)!(n-k)!k}{(n-r-k+1)!n!}$$

The probability that a selector chooses member *x*, when *k* members are sampled, and a new inferior member joins the club, is:

(2)
$$\frac{(n-r+1)!(n-k+1)!k}{(n-r-k+2)!(n+1)!}.$$

Thus, the entrance of a new inferior agent increases the selection probability of member x if:

(3)
$$\frac{(n-r+1)!(n-k+1)!k}{(n-r-k+2)!(n+1)!} > \frac{(n-r)!(n-k)!k}{(n-r-k+1)!n!}$$

or, by rearranging (3), if:

(4)
$$r > \frac{n+1}{k}.$$

Furthermore, members ranked in the last k-1 positions (the last k-2 positions when an inferior member is added) have zero selection probability. Since ranking in one of the last k positions is equal to ranking in position (n-k+1) or onward (for example, ranking in one of the last 10 positions when there are 100 agents is equal to ranking in the 91st position or onward), members ranked in position (n-k+2) or onward (positions (n-k+3) or onward when an inferior member is admitted) have a zero selection probability. Hence, a necessary condition for the selection probability of a member to be positive when an inferior member joins the club is:

 $(5) \qquad (n-k+3) > r$

By (4) and (5), if k agents are sampled, the admission of a new inferior member to the club increases the selection probability of member x, ranked in position r, iff:

(6)
$$(n-k+3) > r > \frac{n+1}{k}$$
.

The selection probability of member *x* when *k* club members are sampled, and a new *superior* member is admitted to the club, is:

(7)
$$\frac{(n-r)!(n-k+1)!k}{(n-r-k+1)!(n+1)!}$$

There is no positive k such that the expression in (7) is larger than the expression in (1), hence a member of quality r^* decreases the selection probability of all the members that are ranked lower than r^* . Combining this result with (6), we get that entry of a new agent ranked r^* increases the selection probability of member x iff:

$$\min[(n-k+3), r^*] > r > \frac{n+1}{k}. \blacksquare$$

Proposition 2: Suppose that *k* club members are sampled. The maximal selection probability of a member ranked *r* is obtained when the number of club members is equal to $n_r = kr$.

Proof:

When n = rk or n = (rk-1), by (1), the selection probability of a member of quality r is the same and equal to $\frac{(n-r)!(n-k)!k}{(n-r-k+1)!n!} = \frac{(rk-r)!(rk-k)!k}{(rk-r-k+1)!(rk)!}.$

The selection probability of a member with quality r when there are (rk+1) members is smaller than the selection probability when there are (rk) members. By induction, it can be verified that the selection probability of a member with quality r when there are (rk+a+1) members is smaller than the selection probability when there are (rk+a) members. The selection probability of a member with quality r when there are (rk-2) members is smaller than the selection probability when there are (rk-2) members is smaller than the selection probability when there are (rk-1) members. By induction, it can be verified that the selection probability of a member with quality r when there are (rk-a-1) members is smaller than the selection probability of a member with quality r when there are (rk-a-1) members. Hence, the maximal selection probability of a club member ranked r is obtained when there are (rk) or (rk-1) members, which completes the proof.

Proposition 3: A self-governing CRSC that applies a simple majority rule to admit members is either stable or decaying. Assuming that the selector samples $k \ge 3$ agents, a sufficient condition for decay is that $k \le 0.5n$.

Proof:

The club faces a dichotomous choice; whether or not to approve the admission of a prospective entrant. An entrant with above-average quality is rejected by:

- 1. All the members of lower quality, in particular, all the members of equal or less than average quality.
- 2. ((n+1)/k) top quality members.

Hence, the majority of members never approves such an entrant. Only entrants with less than average quality can be approved. To prove that the club is decaying, it

suffices to show that the self-governing club is unstable, approving admittance of inferior candidates. So suppose that $k \ge 3$ and that the club considers the candidacy of an inferior agent and let us prove that $k \le 0.5n$ is a sufficient condition for the majority approval of this candidate. Denote by INT[i] the integer part of any real number [i]. By Proposition 1, when a new inferior agent enters the club, the number of top- quality members whose selection probability does not increase is:

(8)
$$INT[(n+1)/k]$$

and the number of low-quality members whose selection probability does not increase is:

Hence, by (8) and (9), the proportion of members whose selection probability does not increase, when a new inferior candidate is admitted to the club, is:

(10)
$$\frac{\mathrm{INT}[(n+1)/k] + k - 2}{n}$$

If n = 2k, then (10) takes the form:

(11)
$$\frac{\mathrm{INT}[(2k+1)/k] + k - 2}{2k} = \frac{2 + \mathrm{INT}[1/k] + k - 2}{2k} = \frac{1}{2}$$

and the new member is admitted since only half of the members vote against him. For n > 2k then, let n = 2k+a, where *a* is a positive integer. (10) now takes the form:

(12)
$$\frac{\text{INT}[(2k+a+1)/k]+k-2}{2k+a} = \frac{2+\text{INT}[(a+1)/k]+k-2}{2k+a}$$

The new candidate is approved if half or fewer of the members vote against him; that is, if:

(13)
$$\frac{1}{2} \ge \frac{k + \text{INT}[(a+1)/k]}{2k+a}$$

Let b = INT[(a+1)/k]. Then (13) can be rewritten as:

(14)
$$\frac{1}{2} \ge \frac{k+b}{2k+a}$$

and (14) holds iff:

$$(15) a \ge 2b$$

or:

(16)
$$a \ge 2INT[(a+1)/k]$$

and, in particular, (16) holds if:

$$(17) a \ge 2(a+1)/k$$

or

$$(18) k \ge 2(a+1)/a$$

If k = 3, it can be verified that (16) is satisfied for a = 1, and that (18) is satisfied for every positive integer a, a > 1. If $k \ge 4$, (18) is satisfied for every positive integer a. Hence, for $n \ge 2k, k \ge 3$, the majority of the existing club members always approves the admission of a new inferior candidate. We have thus proved that the club is decaying.

Proposition 4: Suppose that the number of existing club members *n* is odd. Then the equilibrium size of a self-governing CRSC that applies a simple majority rule is k(n+1)/2. That is, when $k \ge 3$ the *n* existing club members approve the admission of $\{k(n+1)/2 - n\}$ new inferior low-quality members.

Proof:

By Proposition 2, $n_r = kr$ is the most preferred club size of a member ranked *r*. The distribution of these most preferred sizes satisfies the single-peakedness property (see proof of Proposition 3). Hence, by the median voter theorem, the most preferred club size of the median voter k(n+1)/2 is the equilibrium club size under simple majority rule (the proposal k(n+1)/2 is a Condorcet winner: a proposal that defeats any other alternative club size by a simple majority).

Proposition 5: Suppose that the dissemination of information on the quality of the club members positively affects the selector's fixed sample size k. Then a self-governing CRSC that applies a simple majority rule will not disseminate information on its quality.

Proof:

By Proposition 2, $k_r = \frac{n}{r}$ is the most preferred sample size of a member ranked r. The distribution of these most preferred sizes satisfies the single-peakedness property. Hence, by the median voter theorem, the most preferred fixed sample size of the median voter $\frac{2n}{n+1}$ is the equilibrium fixed sample size under simple majority rule.

This means that $k^*=1$ or $k^*=2$ is the voting equilibrium, because it yields the same selection probability to the median voter. Since, by assumption, dissemination of information on the quality of the club members positively affects the actual fixed sample size k, the median voter prefers that no information on the club members' quality is disclosed and this will be the voting equilibrium outcome.

Proposition 6: Suppose that a competitive self-governing CRSC applies a simple majority rule when making marginal or non-marginal admittance decisions or when making a decision on disclosure of information about the quality of its members. Then the CRSC never makes a decision that is advantageous to the selector.

Proof:

(i) The probability that a selector chooses member x when k members are sampled is given in (1). The expected ranking of the selected member is therefore equal to:

(19)
$$\sum_{r=1}^{n-k+1} r(1) = \sum_{r=1}^{n-k+1} \frac{r(n-r)!(n-k)!k}{(n-r-k+1)!n!} = \frac{n+1}{k+1}$$

The selector can benefit only if the entrant's ranking is lower than $\frac{n+1}{k+1}$. By Proposition 1, if a prospective entrant is of such quality, then all the members of higher quality; members whose ranking is lower than or equal to $\frac{n+1}{k}$, vote against entry and, in particular, all the members whose ranking is smaller than $\frac{n+1}{k+1}$, which is smaller than $\frac{n+1}{k}$, vote against entry. By Proposition 1, all the members of quality lower than that of the entrant vote against him. Since all the existing members vote against entry, there is no democratic voting rule that can benefit the selector when entry decisions are marginal.

(ii) When the club-membership decision is not marginal, by Proposition 4 inferior members are admitted to the club and this necessarily reduces the expected quality of the selected member.

(iii) Proposition 5 directly implies that a decision of the club regarding disclosure of information regarding the quality of its members is not advantageous to the selector.■

Proposition 7: In a decentralized CRSC, the (Nash) equilibrium size of the club is nk. That is; the existing n members approve the admission of (k-1)n inferior low-quality candidates.

Proof:

By Proposition 2, $n_r = kr$ is the most preferred club size of a member ranked *r*. It is straightforward to verify that in such a case, the game associated with decentralized club-size decisions possesses the unique Nash equilibrium $(e_1^*, ..., e_n^*) =$

 $(0,0,\ldots,(k-1)n)$. Hence, $e_1^* + e_2^* + \ldots + e_n^* = (k-1)n$. That is, the (Nash) equilibrium size of the club is nk

Proposition 8: Suppose that the members of a fixed-size CRSC have equal ability to disseminate information on the quality of the club members and that this information positively affects the selector's fixed sample size k. Then in a decentralized CRSC, the (Nash) equilibrium of disseminated information on the club's members' quality is maximal.

Proof:

By Proposition 2, $k_r = \frac{n}{r}$ is the most preferred sample size of a member ranked *r*. It is straightforward to verify that in such a case, the game associated with decentralized decisions on disclosure of information regarding the quality of the club members possesses the unique Nash equilibrium where the highest-quality club member disseminates the maximal possible amount of information and the other members do not disseminate information.

References

- Amegashie, A.J., 2000. Some results on rent-seeking contests with shortlisting, *Public Choice*, 105, 245-253.
- Baik, K.H., 2008, Contests with group-specific public-good prizes, *Social Choice and Welfare*, 30(1), 103-117.
- Baik, K.H., 1999, Rent-seeking firms, consumer groups, and the social costs of monopoly. *Economic Inquiry*, 37(3), 542-554.
- Baik, K.H., Kim, I-G. and Na, S-H., 2001, Bidding for a group-specific public-good prize, *Journal of Public Economics*, 82, 415-429.
- Benesch, C, Frey, B. and Stutzer, A., 2006, TV channels, self control and happiness, Working Paper 301, Institute for Empirical Research in Economics, University of Zurich.
- Congleton, R.D., Hillman, A.L. and Konrad, K.A. (Eds), 2008, Forty Years of Research on Rent Seeking, Springer- Verlag.
- Cornes, R. and Sandler, T., 1996. *The Theory of Externalities, Public Goods, and Club Goods*, 2nd edition. Cambridge: Cambridge University Press.
- Deneckere, R. J. and McAfee, R. P., 1996, Damaged goods, *Journal of Economics & Management Strategy*, 5-(2), 149-174.
- Dyer, O., 2003. Heart surgeons are to be rated according to bypass surgery success, *BMJ*;326: 1053-a. (17 May.).
- Ellingsen, T., 1991. Strategic buyers and the social cost of monopoly, American Economic Review, 81, 648-657.
- Fabella, R.V., 1995. The social cost of rent seeking under countervailing opposition to distortionary transfers, *Journal of Public Economics*, 57, 235-247.
- Gans, J., Smart, M., 1996. Majority voting with single crossing preferences, *Journal* of *Public Economics*, 59, 219-237.
- Gradstein, M. (1995). 'Intensity of competition, entry and entry deterrence in rentseeking contests', *Economics and Politics*, vol. 7, pp. 79-91.
- Hausken, K., 2005, Production and conflict models versus rent seeking models. *Public Choice*, 123 (1-2), 59-93.
- Konrad, K.A., 2008, *Strategy and Dynamics in Contests*, Oxford University Press, forthcoming.

- Nitzan, S., 1994, Modelling rent-seeking contests, *European Journal of Political Economy* 10, 41–60.
- Nti, K.O., 1997, Comparative statics of contests and rent seeking games. *International Economic Review*, 38, 43-59.
- Rubinstein, A., 1998. Models of Bounded Rationality, MIT Press.
- Rubinstein, A., Salant, Y., 2006. A model of choice from lists, *Theoretical Economics*, 1, 3-17.
- Sah, R.K., Stiglitz, J.E., 1986. The architecture of economic systems, hierarchies and polyarchies, *American Economic Review*, 76, 716-727.

Bar-Ilan University Department of Economics WORKING PAPERS

1-01 The Optimal Size for a Minority

Hillel Rapoport and Avi Weiss, January 2001.

2-01 An Application of a Switching Regimes Regression to the Study of Urban Structure

Gershon Alperovich and Joseph Deutsch, January 2001.

3-01 The Kuznets Curve and the Impact of Various Income Sources on the Link Between Inequality and Development

Joseph Deutsch and Jacques Silber, February 2001.

4-01 International Asset Allocation: A New Perspective

Abraham Lioui and Patrice Poncet, February 2001.

5-01 מודל המועדון והקהילה החרדית

יעקב רוזנברג, פברואר 2001.

- 6-01 **Multi-Generation Model of Immigrant Earnings: Theory and Application** Gil S. Epstein and Tikva Lecker, February 2001.
- 7-01 Shattered Rails, Ruined Credit: Financial Fragility and Railroad Operations in the Great Depression

Daniel A. Schiffman, February 2001.

- 8-01 **Cooperation and Competition in a Duopoly R&D Market** Damiano Bruno Silipo and Avi Weiss, March 2001.
- 9-01 A Theory of Immigration Amnesties Gil S. Epstein and Avi Weiss, April 2001.
- 10-01 **Dynamic Asset Pricing With Non-Redundant Forwards** Abraham Lioui and Patrice Poncet, May 2001.
- 11-01 Macroeconomic and Labor Market Impact of Russian Immigration in Israel Sarit Cohen and Chang-Tai Hsieh, May 2001.

Electronic versions of the papers are available at http://www.biu.ac.il/soc/ec/wp/working_papers.html

- 12-01 Network Topology and the Efficiency of Equilibrium Igal Milchtaich, June 2001.
- 13-01 General Equilibrium Pricing of Trading Strategy Risk Abraham Lioui and Patrice Poncet, July 2001.
- 14-01 Social Conformity and Child Labor Shirit Katav-Herz, July 2001.
- 15-01 **Determinants of Railroad Capital Structure, 1830–1885** Daniel A. Schiffman, July 2001.
- 16-01 **Political-Legal Institutions and the Railroad Financing Mix, 1885–1929** Daniel A. Schiffman, September 2001.
- 17-01 Macroeconomic Instability, Migration, and the Option Value of Education Eliakim Katz and Hillel Rapoport, October 2001.
- 18-01 **Property Rights, Theft, and Efficiency: The Biblical Waiver of Fines in the Case of Confessed Theft**

Eliakim Katz and Jacob Rosenberg, November 2001.

19-01 Ethnic Discrimination and the Migration of Skilled Labor

Frédéric Docquier and Hillel Rapoport, December 2001.

1-02 Can Vocational Education Improve the Wages of Minorities and Disadvantaged Groups? The Case of Israel

Shoshana Neuman and Adrian Ziderman, February 2002.

2-02 What Can the Price Gap between Branded and Private Label Products Tell Us about Markups?

Robert Barsky, Mark Bergen, Shantanu Dutta, and Daniel Levy, March 2002.

- 3-02 Holiday Price Rigidity and Cost of Price Adjustment Daniel Levy, Georg Müller, Shantanu Dutta, and Mark Bergen, March 2002.
- 4-02 **Computation of Completely Mixed Equilibrium Payoffs**

Igal Milchtaich, March 2002.

5-02 **Coordination and Critical Mass in a Network Market – An Experimental Evaluation**

Amir Etziony and Avi Weiss, March 2002.

- 6-02 **Inviting Competition to Achieve Critical Mass** Amir Etziony and Avi Weiss, April 2002.
- 7-02 **Credibility, Pre-Production and Inviting Competition in a Network Market** Amir Etziony and Avi Weiss, April 2002.
- 8-02 **Brain Drain and LDCs' Growth: Winners and Losers** Michel Beine, Fréderic Docquier, and Hillel Rapoport, April 2002.
- 9-02 Heterogeneity in Price Rigidity: Evidence from a Case Study Using Micro-Level Data

Daniel Levy, Shantanu Dutta, and Mark Bergen, April 2002.

- 10-02 **Price Flexibility in Channels of Distribution: Evidence from Scanner Data** Shantanu Dutta, Mark Bergen, and Daniel Levy, April 2002.
- 11-02 Acquired Cooperation in Finite-Horizon Dynamic Games Igal Milchtaich and Avi Weiss, April 2002.
- 12-02 **Cointegration in Frequency Domain** Daniel Levy, May 2002.

13-02 Which Voting Rules Elicit Informative Voting?

Ruth Ben-Yashar and Igal Milchtaich, May 2002.

14-02 Fertility, Non-Altruism and Economic Growth: Industrialization in the Nineteenth Century

Elise S. Brezis, October 2002.

15-02 Changes in the Recruitment and Education of the Power Elitesin Twentieth Century Western Democracies

Elise S. Brezis and François Crouzet, November 2002.

- 16-02 **On the Typical Spectral Shape of an Economic Variable** Daniel Levy and Hashem Dezhbakhsh, December 2002.
- 17-02 International Evidence on Output Fluctuation and Shock Persistence Daniel Levy and Hashem Dezhbakhsh, December 2002.
- 1-03 **Topological Conditions for Uniqueness of Equilibrium in Networks** Igal Milchtaich, March 2003.
- 2-03 Is the Feldstein-Horioka Puzzle Really a Puzzle? Daniel Levy, June 2003.

- 3-03 Growth and Convergence across the US: Evidence from County-Level Data Matthew Higgins, Daniel Levy, and Andrew Young, June 2003.
- 4-03 **Economic Growth and Endogenous Intergenerational Altruism** Hillel Rapoport and Jean-Pierre Vidal, June 2003.
- 5-03 Remittances and Inequality: A Dynamic Migration Model

Frédéric Docquier and Hillel Rapoport, June 2003.

6-03 Sigma Convergence Versus Beta Convergence: Evidence from U.S. County-Level Data

Andrew T. Young, Matthew J. Higgins, and Daniel Levy, September 2003.

7-03 Managerial and Customer Costs of Price Adjustment: Direct Evidence from Industrial Markets

Mark J. Zbaracki, Mark Ritson, Daniel Levy, Shantanu Dutta, and Mark Bergen, September 2003.

8-03 First and Second Best Voting Rules in Committees

Ruth Ben-Yashar and Igal Milchtaich, October 2003.

9-03 Shattering the Myth of Costless Price Changes: Emerging Perspectives on Dynamic Pricing

Mark Bergen, Shantanu Dutta, Daniel Levy, Mark Ritson, and Mark J. Zbaracki, November 2003.

1-04 Heterogeneity in Convergence Rates and Income Determination across U.S. States: Evidence from County-Level Data

Andrew T. Young, Matthew J. Higgins, and Daniel Levy, January 2004.

2-04 **"The Real Thing:" Nominal Price Rigidity of the Nickel Coke, 1886-1959** Daniel Levy and Andrew T. Young, February 2004.

3-04 Network Effects and the Dynamics of Migration and Inequality: Theory and Evidence from Mexico

David Mckenzie and Hillel Rapoport, March 2004.

4-04 Migration Selectivity and the Evolution of Spatial Inequality

Ravi Kanbur and Hillel Rapoport, March 2004.

5-04 Many Types of Human Capital and Many Roles in U.S. Growth: Evidence from County-Level Educational Attainment Data

Andrew T. Young, Daniel Levy and Matthew J. Higgins, March 2004.

6-04 When Little Things Mean a Lot: On the Inefficiency of Item Pricing Laws

Mark Bergen, Daniel Levy, Sourav Ray, Paul H. Rubin and Benjamin Zeliger, May 2004.

7-04 Comparative Statics of Altruism and Spite

Igal Milchtaich, June 2004.

8-04 Asymmetric Price Adjustment in the Small: An Implication of Rational Inattention

Daniel Levy, Haipeng (Allan) Chen, Sourav Ray and Mark Bergen, July 2004.

1-05 Private Label Price Rigidity during Holiday Periods

Georg Müller, Mark Bergen, Shantanu Dutta and Daniel Levy, March 2005.

2-05 Asymmetric Wholesale Pricing: Theory and Evidence

Sourav Ray, Haipeng (Allan) Chen, Mark Bergen and Daniel Levy, March 2005.

3-05 Beyond the Cost of Price Adjustment: Investments in Pricing Capital

Mark Zbaracki, Mark Bergen, Shantanu Dutta, Daniel Levy and Mark Ritson, May 2005.

4-05 Explicit Evidence on an Implicit Contract

Andrew T. Young and Daniel Levy, June 2005.

5-05 **Popular Perceptions and Political Economy in the Contrived World of Harry Potter**

Avichai Snir and Daniel Levy, September 2005.

6-05 Growth and Convergence across the US: Evidence from County-Level Data (revised version)

Matthew J. Higgins, Daniel Levy, and Andrew T. Young, September 2005.

1-06 Sigma Convergence Versus Beta Convergence: Evidence from U.S. County-Level Data (revised version)

Andrew T. Young, Matthew J. Higgins, and Daniel Levy, June 2006.

- 2-06 **Price Rigidity and Flexibility: Recent Theoretical Developments** Daniel Levy, September 2006.
- 3-06 The Anatomy of a Price Cut: Discovering Organizational Sources of the Costs of Price Adjustment

Mark J. Zbaracki, Mark Bergen, and Daniel Levy, September 2006.

4-06 Holiday Non-Price Rigidity and Cost of Adjustment

Georg Müller, Mark Bergen, Shantanu Dutta, and Daniel Levy. September 2006.

2008-01 Weighted Congestion Games With Separable Preferences Igal Milchtaich, October 2008.

2008-02 Federal, State, and Local Governments: Evaluating their Separate Roles in US Growth

Andrew T. Young, Daniel Levy, and Matthew J. Higgins, December 2008.

2008-03 **Political Profit and the Invention of Modern Currency**

Dror Goldberg, December 2008.

2008-04 Static Stability in Games

Igal Milchtaich, December 2008.

2008-05 Comparative Statics of Altruism and Spite

Igal Milchtaich, December 2008.

2008-06 Abortion and Human Capital Accumulation: A Contribution to the Understanding of the Gender Gap in Education

Leonid V. Azarnert, December 2008.

2008-07 Involuntary Integration in Public Education, Fertility and Human Capital

Leonid V. Azarnert, December 2008.

2009-01 Inter-Ethnic Redistribution and Human Capital Investments

Leonid V. Azarnert, January 2009.

2009-02 Group Specific Public Goods, Orchestration of Interest Groups and Free Riding

Gil S. Epstein and Yosef Mealem, January 2009.

2009-03 Holiday Price Rigidity and Cost of Price Adjustment

Daniel Levy, Haipeng Chen, Georg Müller, Shantanu Dutta, and Mark Bergen, February 2009.

2009-04 Legal Tender

Dror Goldberg, April 2009.

2009-05 The Tax-Foundation Theory of Fiat Money

Dror Goldberg, April 2009.

- 2009-06 **The Inventions and Diffusion of Hyperinflatable Currency** Dror Goldberg, April 2009.
- 2009-07 **The Rise and Fall of America's First Bank** Dror Goldberg, April 2009.
- 2009-08 Judicial Independence and the Validity of Controverted Elections Raphaël Franck, April 2009.
- 2009-09 **A General Index of Inherent Risk** Adi Schnytzer and Sara Westreich, April 2009.
- 2009-10 Measuring the Extent of Inside Trading in Horse Betting Markets Adi Schnytzer, Martien Lamers and Vasiliki Makropoulou, April 2009.
- 2009-11 The Impact of Insider Trading on Forecasting in a Bookmakers' Horse Betting Market

Adi Schnytzer, Martien Lamers and Vasiliki Makropoulou, April 2009.

- 2009-12 Foreign Aid, Fertility and Population Growth: Evidence from Africa Leonid V. Azarnert, April 2009.
- 2009-13 A Reevaluation of the Role of Family in Immigrants' Labor Market Activity: Evidence from a Comparison of Single and Married Immigrants

Sarit Cohen-Goldner, Chemi Gotlibovski and Nava Kahana, May 2009.

2009-14 The Efficient and Fair Approval of "Multiple-Cost–Single-Benefit" Projects Under Unilateral Information

Nava Kahanaa, Yosef Mealem and Shmuel Nitzan, May 2009.

2009-15 Après nous le Déluge: Fertility and the Intensity of Struggle against Immigration

Leonid V. Azarnert, June 2009.

2009-16 Is Specialization Desirable in Committee Decision Making?

Ruth Ben-Yashar, Winston T.H. Koh and Shmuel Nitzan, June 2009.

2009-17 Framing-Based Choice: A Model of Decision-Making Under Risk Kobi Kriesler and Shmuel Nitzan, June 2009.

2009-18 Demystifying the 'Metric Approach to Social Compromise with the Unanimity Criterion'

Shmuel Nitzan, June 2009.

2009-19 On the Robustness of Brain Gain Estimates

Michel Beine, Frédéric Docquier and Hillel Rapoport, July 2009.

2009-20 Wage Mobility in Israel: The Effect of Sectoral Concentration

Ana Rute Cardoso, Shoshana Neuman and Adrian Ziderman, July 2009.

2009-21 Intermittent Employment: Work Histories of Israeli Men and Women, 1983–1995

Shoshana Neuman and Adrian Ziderman, July 2009.

2009-22 National Aggregates and Individual Disaffiliation: An International Study

Pablo Brañas-Garza, Teresa García-Muñoz and Shoshana Neuman, July 2009.

2009-23 The Big Carrot: High-Stakes Incentives Revisited

Pablo Brañas-Garza, Teresa García-Muñoz and Shoshana Neuman, July 2009.

2009-24 The Why, When and How of Immigration Amnesties

Gil S. Epstein and Avi Weiss, September 2009.

2009-25 Documenting the Brain Drain of «la Crème de la Crème»: Three Case-Studies on International Migration at the Upper Tail of the Education Distribution

Frédéric Docquier and Hillel Rapoport, October 2009.

2009-26 Remittances and the Brain Drain Revisited: The Microdata Show That More Educated Migrants Remit More

Albert Bollard, David McKenzie, Melanie Morten and Hillel Rapoport, October 2009.

2009-27 Implementability of Correlated and Communication Equilibrium Outcomes in Incomplete Information Games

Igal Milchtaich, November 2009.

2010-01 The Ultimatum Game and Expected Utility Maximization – In View of Attachment Theory

Shaul Almakias and Avi Weiss, January 2010.

2010-02 A Model of Fault Allocation in Contract Law – Moving From Dividing Liability to Dividing Costs

Osnat Jacobi and Avi Weiss, January 2010.

2010-03 Coordination and Critical Mass in a Network Market: An Experimental Investigation

Bradley J. Ruffle, Avi Weiss and Amir Etziony, February 2010.

2010-04 Immigration, fertility and human capital: A model of economic decline of the West

Leonid V. Azarnert, April 2010.

2010-05 Is Skilled Immigration Always Good for Growth in the Receiving Economy?

Leonid V. Azarnert, April 2010.

2010-06 The Effect of Limited Search Ability on the Quality of Competitive Rent-Seeking Clubs

Shmuel Nitzan and Kobi Kriesler, April 2010.

2010-07 **Condorcet vs. Borda in Light of a Dual Majoritarian Approach**

Eyal Baharad and Shmuel Nitzan, April 2010.

2010-08 Prize Sharing in Collective Contests

Shmuel Nitzan and Kaoru Ueda, April 2010.